Author’s response to reviews

Title: No difference in sexual behavior of adolescent girls following Human Papilloma Virus vaccination: A case study two districts in Uganda; Nakasongola and Luwero

Authors:

Judith C Aujo (aujocaro@gmail.com)
Sabrina Bakeera-Kitaka (sabrinakitaka@yahoo.co.uk)
Sarah Kiguli (skwalube@yahoo.co.uk)
Florence Mirembe (flomir2002@yahoo.com)

Version: 4 Date: 23 August 2013

Author’s response to reviews:

The changes that have been made to this manuscript/submission include:

• Indicating that the qualitative component of the study adheres to the RATS guidelines on qualitative work in this covering letter.

• Added a statement that ‘The study team adhered to the RATS guidelines on qualitative research during the conduct of this study’ at the bottom of the section on Qualitative study (page 5 of manuscript).

• Added a statement that ‘The questionnaires had been pretested before the commencement of the study, in a pilot study done in Wakiso district, which validated their use’ in the methods section on page 4.

• Highlighting the ‘Conclusions’ section in the manuscript.

• Adding a ‘Competing Interests Section’ after the conclusions.

• Moving the funding statement to the acknowledgement section.

• Submitted the questionnaires/data collection tools used in the study.